Author’s response to reviews

Title: From positive screen to engagement in treatment: a preliminary study of the impact of a new model of care for prisoners with serious mental illness

Authors:
 Krishna Pillai (Krishna.Pillai@waitematadhb.govt.nzKrisha)
 Paul Rouse (p.rouse@auckland.ac.nz)
 Brian McKenna (Brian.McKenna@mh.org.au)
 Jeremy Skipworth (Jeremy.Skipworthi@waitematadhb.govt.nz)
 James Cavney (James.Cavney@waitematadhb.govt.nz)
 Rees Tapsell (Rees.Tapsell@waikatodhb.health.nz)
 Alexander Simpson (Sandy.Simpson@camh.ca)
 Dominic Madell (Dominic.Madell@middlemore.co.nz)

Version: 1 Date: 05 Nov 2015

Author’s response to reviews:

November 4th, 2015.

Associate Editor (Forensic Psychiatry),
BMC Psychiatry.

Dear Professor Kennedy,

Re: Manuscript "From positive screen to engagement in treatment: a preliminary study of the impact of a new model of care for prisoners with serious mental illness” (BPSY-D-15-00196).
Attached is a revised version of the manuscript for your consideration. We have outlined in an attachment our responses to the individual reviewer recommendations as well as highlighting in the manuscript where associated changes have been made. Thank you and the reviewers for your comments which we believe have led to an improved manuscript.

We look forward to the outcome of this re-consideration.

Yours sincerely,

Brian McKenna,
Professor of Mental Health Nursing,
Australian Catholic University.